Author’s response to reviews

Title: The effect of insulin degludec on risk of symptomatic nocturnal hypoglycaemia in adults with type 1 diabetes and high risk of nocturnal severe hypoglycaemia (the HypoDeg trial): study rationale and design

Authors:

Rikke Agesen (rikke.mette.agesen@regionh.dk)
Amra Alibegovic (amra.ciric.alibegovic@regionh.dk)
Henrik Andersen (henrik.ullits.andersen@regionh.dk)
Henning Beck-Nielsen (henning.beck-nielsen@rsyd.dk)
Peter Gustenhoff (pegu@rn.dk)
Troels Hansen (tkh@dadlnet.dk)
Christoffer Hedetoft (cgh@regionsjaelland.dk)
Tonny Jensen (tonny.joran.jensen@regionh.dk)
Claus Juhl (claus.boegh.juhl@rsyd.dk)
Susanne Lerche (susanne.soegaard.lerche@rsyd.dk)
Kirsten Nørgaard (kirsten.noergaard@regionh.dk)
Hans-Henrik Parving (d028916@dadlnet.dk)
Lise Tarnow (litar@regionsjaelland.dk)
Birger Thorsteinsson (birger.thorsteinsson@regionh.dk)
Ulrik Pedersen-Bjergaard (ulrik.pedersen-bjergaard@regionh.dk)

Version: 2 Date: 05 Jul 2019

Author’s response to reviews:
Dear Editor-in-chief,

Thank you for the second review. Enclosed please find the revised version of our manuscript “The effect of insulin degludec on risk of symptomatic nocturnal hypoglycaemia in subjects with type 1 diabetes and high risk of nocturnal severe hypoglycaemia (the HypoDeg trial): study rationale and design”.

We have carried out the essential revisions as requested.

Kind regards,

Rikke Agesen